Author's response to reviews

Title: Sensitivity and specificity of plasma disappearance rate of indocyanine green as a prognostic indicator in acute liver failure

Authors:

Uta Merle (uta_merle@med.uni-heidelberg.de)
Olivia Sieg (olivi@gmx.net)
Wolfgang Stremmel (wolfgang_stremmel@med.uni-heidelberg.de)
Jens Encke (j.encke@ak-neuss.de)
Christoph Eisenbach (christoph_eisenbach@med.uni-heidelberg.de)

Version: 3 Date: 22 November 2009

Author's response to reviews:

Dear BioMed Central Editorial Production Team,
As requested we did several formatting changes:
1. We embedded the table in the text.
2. We removed additional files.
3. We re-formatted the title in sentence case.
4. We renamed 'Introduction' in 'Background'.
5. We renamed 'Materials and methods' in 'Methods'.
6. We removed the numbers of the figures from the image files.

The Trial registration number was already included in the submission system and manuscript.
Unfortunately we were unable to crop the figures as requested and would be grateful, if you could do this for us.
Thanks a lot and kind regards,
Uta Merle